Author's response to reviews

Title: A novel scoring system for prognostic prediction in d-galactosamine/lipopolysaccharide-induced fulminant hepatic failure BALB/c mice

Authors:

bo feng (xyfyfb_1@sina.com)
shengming wu (wsm@proteomics.cn)
sa lv (lvsa@sina.com)
feng liu (liu1116m@sina.com)
hongsong chen (chen2999@sohu.com)
yan gao (gaoyan6384@163.com)
fangting dong (dft@proteomics.cn)
la wei (weilai@pkuph.edu.cn)

Version: 3 Date: 19 November 2009

Author's response to reviews: see over
Dear Professors:

We do appreciate for your instruction and advices. Following these, we changed the authors’ style, added the streets of affiliations, and removed the horizontal lines. Please check it. If you still have some questions on revision, please contact us without any hesitation.

Thank you again and Best wishes

Sincerely yours

Lai WEI. Prof. MD,Ph.D
Bo FENG. Ph.D
Peking University People’s Hospital
Peking University Hepatology Institute